\documentclass[10pt,letterpaper, final]{report} % \documentclass{} required in all files
\usepackage{../preamble/myWEPS} % This is now our single preamble file inclusion (it assumes a “.sty” file extension)

\begin{document}
The management scenario can be selected and accessed for editing from the management input panel on the bottom of the main WEPS screen (\cref{fig:WEPS_main_screen}) and is shown here (\cref{fig:Management_Input_panel})

\begin{figure}[H]
\centering
	\includegraphics[scale=1.75]{images/Interface_Ref/Management_Input_panel.jpg}
\caption{Management input panel on the WEPS main screen.}
\label{fig:Management_Input_panel}
\end{figure}

A management rotation scenario for a WEPS simulation run can be selected from pre-generated WEPS management (template) files and then modified as necessary.
Management rotation scenarios can be selected from a drop-down choice list (\cref{fig:Mgt_selection_dropdown_choice_list})
by using the down arrow box
\img{images/icons/down_arrow.png}
on the far right side of the bottom "Management" panel of the WEPS main screen (\cref{fig:WEPS_main_screen}).
Navigation is performed by clicking on the ‘closed key’ symbol
\img{images/icons/ssurgoopenkey.png}
or select the folder and press the right arrow key to open the folder to display its contents.
To close a folder, click the ‘open key’ symbol
\img{images/icons/ssurgoclosekey.png}
or press the left arrow key 
to quit displaying the folder contents.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.75]{images/Interface_Ref/man_choice_list.jpg}
\caption{Management selection drop down choice list.}
\label{fig:Mgt_selection_dropdown_choice_list}
\end{figure} 

The user can also select a management rotation by clicking on the \textbf{Template} folder
\img{images/icons/foldert.png}
to the left of the management rotation name, which will bring up a standard file chooser window (\cref{fig:Template_Man_FileChooser}).
The user may also select previously used management files stored in the current WEPS Project directory by clicking on the \textbf{Project} folder
\img{images/icons/folderp.png}
button.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.75]{images/Interface_Ref/Template_Man_FileChooser.jpg}
\caption{Template management file chooser window.}
\label{fig:Template_Man_FileChooser}
\end{figure} 

Management rotation files in a WEPS Project directory are usually derivatives
of those selected previously from the Template directory with local “Project-specific” modifications.
Note that the
\img{images/icons/folderp.png}
and
\img{images/icons/foldert.png}
buttons are not displayed by default for the official NRCS configured releases of WEPS.
Therefore, these users can only use the down arrow box
\img{images/icons/down_arrow.png}
to access the drop-down choice list for selecting management rotation scenarios from the main WEPS screen.

Management rotation scenarios for a simulation run can also be created/selected within the rotation editor
which can be opened by using the
\scalerel*{\img{images/icons/manNew.jpg}}{\strut}
button on the left side of the bottom panel of the WEPS main screen (\cref{fig:Management_Input_panel}). 
To open the Management/Crop Rotation Editor for WEPS (MCREW), double click on the
\scalerel*{\img{images/icons/manNew.jpg}}{\strut}
button on the left side of the management box. This will open the MCREW window (\cref{fig:Man_Crop_Rotation_Editor},
which allows the user to view, edit, and save management rotation information.
Details on editing a management rotation file are discussed here: \hl{Editing with the Management/Crop Rotation Editor for WEPS (MCREW)}.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.75]{images/Interface_Ref/MCREW_Window.jpg}
\caption{Management/Crop Rotation Editor for WEPS (MCREW) window.}
\label{fig:Man_Crop_Rotation_Editor}
\end{figure}

 
\end{document}